Author’s response to reviews

Title: Lymphangioleiomyomatosis, multifocal micronodular pneumocyte hyperplasia, and sarcoidosis: more pathological findings in the same chest CT, or a single pathological pathway?

Authors:

Fabiano Di Marco (fabiano.dimarco@unimi.it)
Giuseppina Palumbo (palumbo.ggp@gmail.com)
Silvia Terraneo (silvia.terraneo@unimi.it)
Gianluca Imeri (gianluca.imeri@gmail.com)
Elena Lesma (elena.lesma@unimi.it)
Nicola Sverzellati (nicolasve@tiscali.it)
Angela Peron (angela.peron@unimi.it)
Lorenzo Gualandri (lorenzo.gualandri@asst-santipaolocarlo.it)
Maria Paola Canevini (maria.canevini@asst-santipaolocarlo.it)
Stefano Centanni (stefano.centanni@unimi.it)

Version: 3 Date: 10 Jul 2017

Author’s response to reviews:

Milan, 10th July 2017

To Sanjay Haresh Chotirmall, MD PhD
Editor of BMC Pulmonary Medicine

Dear editor,

We appreciate very much your opinion about the final version of our manuscript and we thank you for your final decision.

Following, we have clarified all your concerns.
Best regards

Fabiano Di Marco, MD, PhD

1. As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Author: I declared I am an Associate Editor in the Competing Interests section.

2. Thank you for providing the patient consent form and the statement in the "Consent for publication" section. Please amend the statement, to explicitly refer to the fact that the patient gave consent to publish their case report (and potentially-identifying information), as well as any accompanying images.

Authors: the "Consent for publication" section has been amended

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors: we uploaded the final clean version of the manuscript.